\begin{thebibliography}{}
\expandafter\ifx\csname natexlab\endcsname\relax\def\natexlab#1{#1}\fi
\providecommand{\url}[1]{\href{#1}{#1}}
\providecommand{\dodoi}[1]{doi:~\href{http://doi.org/#1}{\nolinkurl{#1}}}
\providecommand{\doeprint}[1]{\href{http://ascl.net/#1}{\nolinkurl{http://ascl.net/#1}}}
\providecommand{\doarXiv}[1]{\href{https://arxiv.org/abs/#1}{\nolinkurl{https://arxiv.org/abs/#1}}}

\bibitem[{{Alcaino}(1975)}]{Alcaino1975}
{Alcaino}, G. 1975, \aaps, 21, 15

\bibitem[{{Anderson} {et~al.}(2008){Anderson}, {Sarajedini}, {Bedin}, {King},
  {Piotto}, {Reid}, {Siegel}, {Majewski}, {Paust}, {Aparicio}, {Milone},
  {Chaboyer}, \& {Rosenberg}}]{Anderson2008}
{Anderson}, J., {Sarajedini}, A., {Bedin}, L.~R., {et~al.} 2008, \aj, 135,
  2055, \dodoi{10.1088/0004-6256/135/6/2055}

\bibitem[{Barber {et~al.}(1996)Barber, Dobkin, \& Huhdanpaa}]{Barber1996}
Barber, C.~B., Dobkin, D.~P., \& Huhdanpaa, H. 1996, ACM Transactions on
  Mathematical Software (TOMS), 22, 469

\bibitem[{{Bastian} \& {Lardo}(2015)}]{Bastian2015}
{Bastian}, N., \& {Lardo}, C. 2015, \mnras, 453, 357,
  \dodoi{10.1093/mnras/stv1661}

\bibitem[{Bastian \& Lardo(2018)}]{Bastian2018}
Bastian, N., \& Lardo, C. 2018, Annual Review of Astronomy and Astrophysics,
  56, 83

\bibitem[{Bekki \& Chiba(2002)}]{Bekki2002}
Bekki, K., \& Chiba, M. 2002, The Astrophysical Journal, 566, 245,
  \dodoi{10.1086/337984}

\bibitem[{{Boudreaux} \& {Chaboyer}(2023)}]{Boudreaux2023}
{Boudreaux}, E.~M., \& {Chaboyer}, B.~C. 2023, \apj, 944, 129,
  \dodoi{10.3847/1538-4357/acb685}

\bibitem[{{Boylan-Kolchin}(2018)}]{Boy18}
{Boylan-Kolchin}, M. 2018, \mnras, 1423, \dodoi{10.1093/mnras/sty1490}

\bibitem[{Brodie \& Strader(2006)}]{Bro06}
Brodie, J.~P., \& Strader, J. 2006, Annu. Rev. Astron. Astrophys., 44, 193

\bibitem[{{Brown} {et~al.}(1991){Brown}, {Burkert}, \& {Truran}}]{Brown1991}
{Brown}, J.~H., {Burkert}, A., \& {Truran}, J.~W. 1991, \apj, 376, 115,
  \dodoi{10.1086/170260}

\bibitem[{{Brown} {et~al.}(1995){Brown}, {Burkert}, \& {Truran}}]{Brown1995}
---. 1995, \apj, 440, 666, \dodoi{10.1086/175304}

\bibitem[{{Cadelano} {et~al.}(2022){Cadelano}, {Dalessandro}, {Salaris},
  {Bastian}, {Mucciarelli}, {Saracino}, {Martocchia}, \&
  {Cabrera-Ziri}}]{Cadelano2022}
{Cadelano}, M., {Dalessandro}, E., {Salaris}, M., {et~al.} 2022, \apjl, 924,
  L2, \dodoi{10.3847/2041-8213/ac424a}

\bibitem[{{Carretta}(2006)}]{Carretta2006}
{Carretta}, E. 2006, \aj, 131, 1766, \dodoi{10.1086/499565}

\bibitem[{{Carretta}(2013)}]{Carretta2013}
---. 2013, \aap, 557, A128, \dodoi{10.1051/0004-6361/201322103}

\bibitem[{{Carretta}(2015)}]{Carretta2015}
---. 2015, \apj, 810, 148, \dodoi{10.1088/0004-637X/810/2/148}

\bibitem[{{Carretta} {et~al.}(2004){Carretta}, {Bragaglia}, \&
  {Cacciari}}]{Carretta2004}
{Carretta}, E., {Bragaglia}, A., \& {Cacciari}, C. 2004, \apjl, 610, L25,
  \dodoi{10.1086/423034}

\bibitem[{Carretta {et~al.}(2010)Carretta, Bragaglia, Gratton, Recio-Blanco,
  Lucatello, D'Orazi, \& Cassisi}]{Carretta2010}
Carretta, E., Bragaglia, A., Gratton, R.~G., {et~al.} 2010, Astronomy \&
  Astrophysics, 516, A55

\bibitem[{{Colgan} {et~al.}(2016){Colgan}, {Kilcrease}, {Magee}, { Sherrill},
  {Abdallah}, {Hakel}, {Fontes}, {Guzik}, \& {Mussack}}]{Colgan2016}
{Colgan}, J., {Kilcrease}, D.~P., {Magee}, N.~H., {et~al.} 2016, in APS Meeting
  Abstracts, Vol. 2016, APS Division of Atomic, Molecular and Optical Physics
  Meeting Abstracts, D1.008

\bibitem[{{de Mink} {et~al.}(2009){de Mink}, {Pols}, {Langer}, \&
  {Izzard}}]{deMink2009}
{de Mink}, S.~E., {Pols}, O.~R., {Langer}, N., \& {Izzard}, R.~G. 2009, \aap,
  507, L1, \dodoi{10.1051/0004-6361/200913205}

\bibitem[{{Decressin} {et~al.}(2007){Decressin}, {Meynet}, {Charbonnel},
  {Prantzos}, \& {Ekstr{\"o}m}}]{Decressin2007}
{Decressin}, T., {Meynet}, G., {Charbonnel}, C., {Prantzos}, N., \&
  {Ekstr{\"o}m}, S. 2007, \aap, 464, 1029, \dodoi{10.1051/0004-6361:20066013}

\bibitem[{{Denissenkov} \& {Hartwick}(2014)}]{Denissenkov2014}
{Denissenkov}, P.~A., \& {Hartwick}, F.~D.~A. 2014, \mnras, 437, L21,
  \dodoi{10.1093/mnrasl/slt133}

\bibitem[{{D'Ercole} {et~al.}(2010){D'Ercole}, {D'Antona}, {Ventura},
  {Vesperini}, \& {McMillan}}]{DErcole2010}
{D'Ercole}, A., {D'Antona}, F., {Ventura}, P., {Vesperini}, E., \& {McMillan},
  S. L.~W. 2010, \mnras, 407, 854, \dodoi{10.1111/j.1365-2966.2010.16996.x}

\bibitem[{{D'Ercole} {et~al.}(2008){D'Ercole}, {Vesperini}, {D'Antona},
  {McMillan}, \& {Recchi}}]{DErcole2008}
{D'Ercole}, A., {Vesperini}, E., {D'Antona}, F., {McMillan}, S. L.~W., \&
  {Recchi}, S. 2008, \mnras, 391, 825, \dodoi{10.1111/j.1365-2966.2008.13915.x}

\bibitem[{{Dotter}(2016)}]{Dotter2016}
{Dotter}, A. 2016, \apjs, 222, 8, \dodoi{10.3847/0067-0049/222/1/8}

\bibitem[{Dotter {et~al.}(2008)Dotter, Chaboyer, Jevremovi{\'c}, Kostov, Baron,
  \& Ferguson}]{Dotter2008}
Dotter, A., Chaboyer, B., Jevremovi{\'c}, D., {et~al.} 2008, The Astrophysical
  Journal Supplement Series, 178, 89

\bibitem[{{Dotter} {et~al.}(2015){Dotter}, {Ferguson}, {Conroy}, { Milone},
  {Marino}, \& {Yong}}]{Dotter2015}
{Dotter}, A., {Ferguson}, J.~W., {Conroy}, C., {et~al.} 2015, \mnras, 446,
  1641, \dodoi{10.1093/mnras/stu2170}

\bibitem[{Ferguson(1973)}]{Ferguson1973}
Ferguson, T.~S. 1973, The annals of statistics, 209

\bibitem[{{Gratton} {et~al.}(2004){Gratton}, {Sneden}, \&
  {Carretta}}]{Gratton2004}
{Gratton}, R., {Sneden}, C., \& {Carretta}, E. 2004, \araa, 42, 385,
  \dodoi{10.1146/annurev.astro.42.053102.133945}

\bibitem[{{Gratton} {et~al.}(2012){Gratton}, {Carretta}, \&
  {Bragaglia}}]{Gratton2012}
{Gratton}, R.~G., {Carretta}, E., \& {Bragaglia}, A. 2012, Astronomy and
  Astrophysics Reviews, 20, 50, \dodoi{10.1007/s00159-012-0050-3}

\bibitem[{{Gratton} {et~al.}(2011){Gratton}, {Lucatello}, {Carretta},
  {Bragaglia}, {D'Orazi}, \& {Momany}}]{Gratton2011}
{Gratton}, R.~G., {Lucatello}, S., {Carretta}, E., {et~al.} 2011, \aap, 534,
  A123, \dodoi{10.1051/0004-6361/201117690}

\bibitem[{{Grevesse} {et~al.}(2007){Grevesse}, {Asplund}, \&
  {Sauval}}]{Grevesse2007}
{Grevesse}, N., {Asplund}, M., \& {Sauval}, A.~J. 2007, \ssr, 130, 105,
  \dodoi{10.1007/s11214-007-9173-7}

\bibitem[{{Gustafsson} {et~al.}(2008){Gustafsson}, {Edvardsson}, {Eriksson},
  {J{\o} rgensen}, {Nordlund}, \& {Plez}}]{Gustafsson2008}
{Gustafsson}, B., {Edvardsson}, B., {Eriksson}, K., {et~al.} 2008, \aap, 486,
  951, \dodoi{10.1051/0004-6361:200809724}

\bibitem[{{Hong} {et~al.}(2021){Hong}, {Lim}, {Chung}, {Kim}, {Han}, \&
  {Lee}}]{Hong2021}
{Hong}, S., {Lim}, D., {Chung}, C., {et~al.} 2021, \aj, 162, 130,
  \dodoi{10.3847/1538-3881/ac0ce6}

\bibitem[{{Hudson} \& {Robison}(2018)}]{Hud18}
{Hudson}, M.~J., \& {Robison}, B. 2018, Monthly Notices of the Royal
  Astronomical Society, 477, 3869, \dodoi{10.1093/mnras/sty844}

\bibitem[{{Husser} {et~al.}(2013){Husser}, {Wende-von Berg}, {Dreizler}, {
  Homeier}, {Reiners}, {Barman}, \& {Hauschildt}}]{Husser2013}
{Husser}, T.~O., {Wende-von Berg}, S., {Dreizler}, S., {et~al.} 2013, \aap,
  553, A6, \dodoi{10.1051/0004-6361/201219058}

\bibitem[{{Jord{\'a}n} {et~al.}(2002){Jord{\'a}n}, {C{\^o}t{\'e}}, {West}, \&
  {Marzke}}]{Jordan2002}
{Jord{\'a}n}, A., {C{\^o}t{\'e}}, P., {West}, M.~J., \& {Marzke}, R.~O. 2002,
  \apjl, 576, L113, \dodoi{10.1086/343759}

\bibitem[{{Kalirai} \& {Richer}(2010)}]{Kalirai2010}
{Kalirai}, J.~S., \& {Richer}, H.~B. 2010, Philosophical Transactions of the
  Royal Society of London Series A, 368, 755, \dodoi{10.1098/rsta.2009.0257}

\bibitem[{Kostogryz {et~al.}(2023)Kostogryz, Shapiro, Witzke, Grant, Wakeford,
  Stevenson, Solanki, \& Gizon}]{Kostogryz2023}
Kostogryz, N., Shapiro, A.~I., Witzke, V., {et~al.} 2023, Research Notes of the
  AAS, 7, 39, \dodoi{10.3847/2515-5172/acc180}

\bibitem[{Kravtsov \& Gnedin(2005)}]{Kra05}
Kravtsov, A.~V., \& Gnedin, O.~Y. 2005, The Astrophysical Journal, 623, 650

\bibitem[{Kurucz(1993)}]{Kurucz1993}
Kurucz, R.-L. 1993, Kurucz CD-Rom, 13

\bibitem[{{Latour} {et~al.}(2019){Latour}, {Husser}, {Giesers}, {Kamann},
  {G{\"o}ttgens}, {Dreizler}, {Brinchmann}, { Bastian}, {Wendt}, {Weilbacher},
  \& {Molinski}}]{Latour2019}
{Latour}, M., {Husser}, T.~O., {Giesers}, B., {et~al.} 2019, \aap, 631, A14,
  \dodoi{10.1051/0004-6361/201936242}

\bibitem[{{Marigo} \& {Aringer}(2009)}]{Marigo2009}
{Marigo}, P., \& {Aringer}, B. 2009, \aap, 508, 1539,
  \dodoi{10.1051/0004-6361/200912598}

\bibitem[{{Marigo} {et~al.}(2022){Marigo}, {Aringer}, {Girardi}, \& {
  Bressan}}]{Marigo2022}
{Marigo}, P., {Aringer}, B., {Girardi}, L., \& { Bressan}, A. 2022, \apj, 940,
  129, \dodoi{10.3847/1538-4357/ac9b40}

\bibitem[{Marino {et~al.}(2015)Marino, Milone, Karakas, Casagrande, Yong,
  Shingles, Da~Costa, Norris, Stetson, Lind, Asplund, Collet, Jerjen, Sbordone,
  Aparicio, \& Cassisi}]{Marino2015}
Marino, A.~F., Milone, A.~P., Karakas, A.~I., {et~al.} 2015, Monthly Notices of
  the Royal Astronomical Society, 450, 815, \dodoi{10.1093/mnras/stv420}

\bibitem[{Martocchia {et~al.}(2019)Martocchia, Dalessandro, Lardo,
  Cabrera-Ziri, Bastian, Kozhurina-Platais, Salaris, Chantereau, Geisler,
  Hilker, Kacharov, Larsen, Mucciarelli, Niederhofer, Platais, \&
  Usher}]{Martocchia2019}
Martocchia, S., Dalessandro, E., Lardo, C., {et~al.} 2019, Monthly Notices of
  the Royal Astronomical Society, 487, 5324, \dodoi{10.1093/mnras/stz1596}

\bibitem[{{Milone} {et~al.}(2012){Milone}, {Piotto}, {Bedin}, {King},
  {Anderson}, {Marino}, {Bellini}, { Gratton}, {Renzini}, {Stetson}, {Cassisi},
  {Aparicio}, {Bragaglia}, {Carretta}, { D'Antona}, {Di Criscienzo},
  {Lucatello}, { Monelli}, \& {Pietrinferni}}]{Milone2012}
{Milone}, A.~P., {Piotto}, G., {Bedin}, L.~R., {et~al.} 2012, \apj, 744, 58,
  \dodoi{10.1088/0004-637X/744/1/58}

\bibitem[{{Milone} {et~al.}(2015{\natexlab{a}}){Milone}, {Marino}, {Piotto},
  {Renzini}, {Bedin}, {Anderson}, {Cassisi}, { D'Antona}, {Bellini}, {Jerjen},
  {Pietrinferni}, \& {Ventura}}]{Milone2015}
{Milone}, A.~P., {Marino}, A.~F., {Piotto}, G., {et~al.} 2015{\natexlab{a}},
  \apj, 808, 51, \dodoi{10.1088/0004-637X/808/1/51}

\bibitem[{{Milone} {et~al.}(2015{\natexlab{b}}){Milone}, {Marino}, {Piotto},
  {Bedin}, {Anderson}, {Renzini}, {King}, {Bellini}, {Brown}, {Cassisi},
  {D'Antona}, {Jerjen}, {Nardiello}, {Salaris}, {Marel}, {Vesperini}, {Yong},
  {Aparicio}, {Sarajedini}, \& {Zoccali}}]{Milone2015b}
---. 2015{\natexlab{b}}, \mnras, 447, 927, \dodoi{10.1093/mnras/stu2446}

\bibitem[{{Milone} {et~al.}(2017){Milone}, {Piotto}, {Renzini}, {Marino},
  {Bedin}, {Vesperini}, {D'Antona}, {Nardiello}, {Anderson}, {King}, {Yong},
  {Bellini}, {Aparicio}, {Barbuy}, {Brown}, {Cassisi}, {Ortolani}, {Salaris},
  {Sarajedini}, \& {van der Marel}}]{Milone2017}
{Milone}, A.~P., {Piotto}, G., {Renzini}, A., {et~al.} 2017, \mnras, 464, 3636,
  \dodoi{10.1093/mnras/stw2531}

\bibitem[{{Milone} {et~al.}(2018){Milone}, {Marino}, {Renzini}, {D'Antona},
  {Anderson}, {Barbuy}, {Bedin}, {Bellini}, {Brown}, {Cassisi}, {Cordoni},
  {Lagioia}, {Nardiello}, {Ortolani}, {Piotto}, {Sarajedini}, {Tailo}, {van der
  Marel}, \& {Vesperini}}]{Milone2018}
{Milone}, A.~P., {Marino}, A.~F., {Renzini}, A., {et~al.} 2018, \mnras, 481,
  5098, \dodoi{10.1093/mnras/sty2573}

\bibitem[{{Pasquato} \& {Milone}(2019)}]{Pasquato2019}
{Pasquato}, M., \& {Milone}, A. 2019, arXiv e-prints, arXiv:1906.04983,
  \dodoi{10.48550/arXiv.1906.04983}

\bibitem[{Pedregosa {et~al.}(2011)Pedregosa, Varoquaux, Gramfort, Michel,
  Thirion, Grisel, Blondel, Prettenhofer, Weiss, Dubourg, Vanderplas, Passos,
  Cournapeau, Brucher, Perrot, \& Duchesnay}]{scikit-learn}
Pedregosa, F., Varoquaux, G., Gramfort, A., {et~al.} 2011, Journal of Machine
  Learning Research, 12, 2825

\bibitem[{{Peebles} \& {Dicke}(1968)}]{Peebles1968}
{Peebles}, P.~J.~E., \& {Dicke}, R.~H. 1968, \apj, 154, 891,
  \dodoi{10.1086/149811}

\bibitem[{Peng {et~al.}(2011)Peng, Ferguson, Goudfrooij, Hammer, Lucey, Marzke,
  Puzia, Carter, Balcells, Bridges, {et~al.}}]{Pen11}
Peng, E.~W., Ferguson, H.~C., Goudfrooij, P., {et~al.} 2011, The Astrophysical
  Journal, 730, 23

\bibitem[{Piotto(2018)}]{HUGS}
Piotto, G. 2018, HST UV Globular Cluster Survey ("HUGS"),  STScI/MAST,
  \dodoi{10.17909/T9810F}

\bibitem[{{Piotto} {et~al.}(2007){Piotto}, {Bedin}, {Anderson}, {King},
  {Cassisi}, {Milone}, {Villanova}, { Pietrinferni}, \& {Renzini}}]{Piotto2007}
{Piotto}, G., {Bedin}, L.~R., {Anderson}, J., {et~al.} 2007, The Astrophysical
  Journal Letters, 661, L53, \dodoi{10.1086/518503}

\bibitem[{{Piotto} {et~al.}(2015){Piotto}, {Milone}, {Bedin}, {Anderson},
  {King}, {Marino}, {Nardiello}, { Aparicio}, {Barbuy}, {Bellini}, {Brown},
  {Cassisi}, {Cool}, {Cunial}, {Dalessandro }, {D'Antona}, {Ferraro},
  {Hidalgo}, { Lanzoni}, {Monelli}, {Ortolani}, {Renzini}, {Salaris},
  {Sarajedini}, {van der Marel}, {Vesperini}, \& {Zoccali}}]{Piotto2015}
{Piotto}, G., {Milone}, A.~P., {Bedin}, L.~R., {et~al.} 2015, \aj, 149, 91,
  \dodoi{10.1088/0004-6256/149/3/91}

\bibitem[{{Plez}(2008)}]{Plez2008}
{Plez}, B. 2008, Physica Scripta Volume T, 133, 014003,
  \dodoi{10.1088/0031-8949/2008/T133/014003}

\bibitem[{{Prantzos} {et~al.}(2007){Prantzos}, {Charbonnel}, \&
  {Iliadis}}]{Prantzos2007}
{Prantzos}, N., {Charbonnel}, C., \& {Iliadis}, C. 2007, \aap, 470, 179,
  \dodoi{10.1051/0004-6361:20077205}

\bibitem[{Renzini(2008)}]{Renzini2008}
Renzini, A. 2008, Monthly Notices of the Royal Astronomical Society, 391, 354,
  \dodoi{10.1111/j.1365-2966.2008.13892.x}

\bibitem[{{Richer} {et~al.}(1991){Richer}, {Fahlman}, {Buonanno}, {Fusi Pecci},
  {Searle}, \& {Thompson}}]{Richer1991}
{Richer}, H.~B., {Fahlman}, G.~G., {Buonanno}, R., {et~al.} 1991, \apj, 381,
  147, \dodoi{10.1086/170637}

\bibitem[{Rousseeuw(1987)}]{ROUSSEEUW198753}
Rousseeuw, P.~J. 1987, Journal of Computational and Applied Mathematics, 20,
  53, \dodoi{https://doi.org/10.1016/0377-0427(87)90125-7}

\bibitem[{Salaris \& Cassisi(2005)}]{Salaris2005}
Salaris, M., \& Cassisi, S. 2005, Evolution of stars and stellar populations
  (John Wiley \& Sons)

\bibitem[{{Sandage}(1953)}]{Sandage1953}
{Sandage}, A.~R. 1953, \aj, 58, 61, \dodoi{10.1086/106822}

\bibitem[{Shahapure \& Nicholas(2020)}]{Shahapure2020}
Shahapure, K.~R., \& Nicholas, C. 2020, in 2020 IEEE 7th International
  Conference on Data Science and Advanced Analytics (DSAA), 747--748,
  \dodoi{10.1109/DSAA49011.2020.00096}

\bibitem[{Smith(1987)}]{Smith1987}
Smith, G.~H. 1987, Publications of the Astronomical Society of the Pacific, 99,
  67, \dodoi{10.1086/131958}

\bibitem[{Sneden {et~al.}(1992)Sneden, Kraft, Prosser, \& Langer}]{Sneden1992}
Sneden, C., Kraft, R.~P., Prosser, C.~F., \& Langer, G. 1992, The Astronomical
  Journal, 104, 2121

\bibitem[{{Suntzeff} \& {Kraft}(1996)}]{Suntzeff1996}
{Suntzeff}, N.~B., \& {Kraft}, R.~P. 1996, \aj, 111, 1913,
  \dodoi{10.1086/117930}

\bibitem[{{Valle} {et~al.}(2022){Valle}, {Dell'Omodarme}, \&
  {Tognelli}}]{Valle2022}
{Valle}, G., {Dell'Omodarme}, M., \& {Tognelli}, E. 2022, \aap, 658, A141,
  \dodoi{10.1051/0004-6361/202142454}

\bibitem[{van~den Bergh(2010)}]{Vanderburg2010}
van~den Bergh, S. 2010, The Astronomical Journal, 140, 1043,
  \dodoi{10.1088/0004-6256/140/4/1043}

\bibitem[{{Ventura} \& {D'Antona}(2009)}]{Ventura2009}
{Ventura}, P., \& {D'Antona}, F. 2009, \aap, 499, 835,
  \dodoi{10.1051/0004-6361/200811139}

\bibitem[{{Ventura} {et~al.}(2001){Ventura}, {D'Antona}, {Mazzitelli}, \&
  {Gratton}}]{Ventura2001}
{Ventura}, P., {D'Antona}, F., {Mazzitelli}, I., \& {Gratton}, R. 2001, \apjl,
  550, L65, \dodoi{10.1086/319496}

\end{thebibliography}

\section{Stellar Models}\label{sec:modeling}
We use the Dartmouth Stellar Evolution Program \citep[DSEP, ][]{Dotter2008} to
generate stellar models. DSEP is a one-dimensional stellar
evolution code which includes a mixing length model of convection,
gravitational settling, and diffusion. Using the solar composition presented in
\citep{Grevesse2007} (GAS07), MARCS model atmosphers, OPLIB high temperature
opacities, and AESOPUS 2.0 low temperautre opacities we find a solar calibrated
mixing length parameter, $\alpha_{MLT}$, of $\alpha_{MLT} = 1.901$.

We use DSEP to evolve stellar models ranging in mass from 0.3 to 2.0 solar
masses from the fully convective pre-main sequence to the tip of the red giant
branch. Below 0.7 $M_{\odot}$ we evolve a model every 0.03 $M_{\odot}$ and
above 0.7 $M_{\odot}$ we evolve a model every 0.05 $M_{\odot}$. We evolve
models over a grid of mixing length parameters from $\alpha_{MLT} = 1.0$ to
$\alpha_{MLT} = 2.0$ in steps of 0.1. For each mixing length, a grid of models
and isochrones were calculated, with chemical compositions consistent with
Populations A and E (see Table \ref{tab:comp}) and a range of helium abundances
(Y=0.24, 0.27, 0.30, 0.33, 0.36, and 0.39). In total,144 sets of isochrones,
each with a unique composition and mixing length were calculated. Each model is
evolved in DSEP with typical numeric tolerences of one part in $10^{7}$. Each
model is allowed a maximum time step of 50 Myr. 

\begin{deluxetable}{c|cc||c|cc}\label{tab:comp}

%% Keep a portrait orientation

%% Over-ride the default font size
%% Use Default (12pt)

%% Use \tablewidth{?pt} to over-ride the default table width.
%% If you are unhappy with the default look at the end of the
%% *.log file to see what the default was set at before adjusting
%% this value.

%% This is the title of the table.
\tablecaption{Population Composition}

%% This command over-rides LaTeX's natural table count
%% and replaces it with this number.  LaTeX will increment 
%% all other tables after this table based on this number
\tablenum{1}

%% The \tablehead gives provides the column headers.  It
%% is currently set up so that the column labels are on the
%% top line and the units surrounded by ()s are in the 
%% bottom line.  You may add more header information by writing
%% another line between these lines. For each column that requries
%% extra information be sure to include a \colhead{text} command
%% and remember to end any extra lines with \\ and include the 
%% correct number of &s.
  \tablehead{\colhead{Element} & \colhead{Pop A} & \colhead{Pop E} & \colhead{Element} & \colhead{Pop A} & \colhead{Pop E} 
} 

%% All data must appear between the \startdata and \enddata commands
\startdata
Li & -0.08 & --- & In & -1.46 & --- \\
Be & 0.25 & --- & Sn & -0.22 & --- \\
B & 1.57 & --- & Sb & -1.25 & --- \\
C & 6.87 & 5.91 & Te & -0.08 & --- \\
N & 6.42 & 6.69 & I & -0.71 & --- \\
O & 7.87 & 6.91 & Xe & -0.02 & --- \\
F & 3.43 & --- & Cs & -1.18 & --- \\
Ne & 7.12 & 6.7 & Ba & 1.05 & --- \\
Na & 5.11 & 5.7 & La & -0.03 & --- \\
Mg & 6.86 & 6.42 & Ce & 0.45 & --- \\
Al & 5.21 & 6.61 & Pr & -1.54 & --- \\
Si & 6.65 & 6.77 & Nd & 0.29 & --- \\
P & 4.28 & --- & Pm & -99.0 & --- \\
S & 6.31 & 5.89 & Sm & -1.3 & --- \\
Cl & -1.13 & 4.37 & Eu & -0.61 & --- \\
Ar & 5.59 & 5.17 & Gd & -1.19 & --- \\
K & 3.9 & --- & Tb & -1.96 & --- \\
Ca & 5.21 & --- & Dy & -1.16 & --- \\
Sc & 2.02 & --- & Ho & -1.78 & --- \\
Ti & 3.82 & --- & Er & -1.34 & --- \\
V & 2.8 & --- & Tm & -2.16 & --- \\
Cr & 4.51 & --- & Yb & -1.42 & --- \\
Mn & 4.3 & --- & Lu & -2.16 & --- \\
Fe & 6.37 & --- & Hf & -1.41 & --- \\
Co & 3.86 & --- & Ta & -2.38 & --- \\
Ni & 5.09 & --- & W & -1.41 & --- \\
Cu & 3.06 & --- & Re & -2.0 & --- \\
Zn & 2.3 & --- & Os & -0.86 & --- \\
Ga & 0.78 & --- & Ir & -0.88 & --- \\
Ge & 1.39 & --- & Pt & -0.64 & --- \\
As & 0.04 & --- & Au & -1.34 & --- \\
Se & 1.08 & --- & Hg & -1.09 & --- \\
Br & 0.28 & --- & Tl & -1.36 & --- \\
Kr & 0.99 & --- & Pb & -0.51 & --- \\
Rb & 0.26 & --- & Bi & -1.61 & --- \\
Sr & 0.61 & --- & Po & -99.0 & --- \\
Y & 1.08 & --- & At & -99.0 & --- \\
Zr & 1.45 & --- & Rn & -99.0 & --- \\
Nb & -0.8 & --- & Fr & -99.0 & --- \\
Mo & -0.38 & --- & Ra & -99.0 & --- \\
Tc & -99.0 & --- & Ac & -99.0 & --- \\
Ru & -0.51 & --- & Th & -2.2 & --- \\
Rh & -1.35 & --- & Pa & -99.0 & --- \\
Pd & -0.69 & --- & U & -2.8 & --- \\
\enddata

%% Include any \tablenotetext{key}{text}, \tablerefs{ref list},
%% or \tablecomments{text} between the \enddata and 
%% \end{deluxetable} commands

%% General table comment marker
\tablecomments{Relative Metal composition used where a(H) = 12. Where the relative composition is the the same for both populations A and E it is only listed in the population A colum for the sake of visual clarity.}

%% General table references marker
\tablerefs{\citet{Milone2015}}

\end{deluxetable}

For each combination of population, $Y$, and $\alpha_{MLT}$ we use the
isochrone generation code first presented in \citet{Dotter2016} to generate a
grid of isochrones. The isochrone generation code identified equivalent
evolutionary points (EEPs) over a series of masses and interpolates between
them. The grid of isochrones generated for this work is avalible as a digital
supplement to this paper. Given the complexity of the parameter space when
fitting multiple populations along with the recent warnings in the liteerature
regarding overfitting datasets \citep[e.g. ][]{Valle2022} we want to develop a
more objective way of fitting isochrones to photometry than if we were to mark
median ridge line positions by hand.

